Author's response to reviews

Title: RAS mutations in early age leukaemia modulated by NQO1 rs1800566 (C609T) are associated with second-hand smoking exposures

Authors:

Francianne G Andrade (fandrade@inca.gov.br)
Juliana M Furtado-Silva (julianamfs@yahoo.com.br)
Bruno AA Goncalves (baquiar@inca.gov.br)
Luiz C Thuler (lthuler@inca.gov.br)
Thayana C Barbosa (tbarbosa@inca.gov.br)
Mariana Emerenciano (memerenciano@inca.gov.br)
Andre Siqueira (amsiqueira@gmail.com)
Maria S. Pombo-de-Oliveira (mpombo@inca.gov.br)

Version: 6 Date: 14 February 2014

Author's response to reviews: see over
Dear Editor,

Thank you to the decision to accept our manuscript entitled *RAS mutations in early age leukaemia modulated by NQO1 rs1800566(C609T) are associated with second-hand smoking exposures*. I am replying with the minor amendment as requested:

**Study Population and samples.**

All co-participants approved the study according to Brazilian National Research and Ethics Committees (CONEP) followed by the Instituto Nacional de Câncer Research and Ethics Committees (CEP) under the registry: (CEP #005/06, Multi-institutional Study of Infant Leukemia: Contribution of Immunomolecular Markers in Distinguishing Different Etiopathogenic Factors); and CEP #024/10-CONEP #707/2010: Genomic study of infant leukemia with MLL rearrangements). Written consent was obtained from the parents for both polymorphisms and maternal questionnaire procedures.